Author's response to reviews

Title: Sociodemographic and occupational risk factors associated with the development of different burnout types: the cross-sectional University of Zaragoza study

Authors:

Jesus Montero-Marin (jmontero@unizar.es)
Javier Garcia-Campayo (jgarcamp@arrakis.es)
Marta Fajó-Pascual (mfajo@unizar.es)
Jose Miguel Carrasco (jmcarrasco.iacs@aragon.es)
Santiago Gascón (sgascon@unizar.es)
Margarita Gili (mgili@uib.es)
Fermin Mayoral (fermin.mayoral.sspa@juntadeandalucia.es)

Version: 4 Date: 14 March 2011

Author's response to reviews:

We have carried out the modifications suggested by the editors

The authors